Author's response to reviews

Title: Effect of delayed CNI-based immunosuppression with Advagraf(R) on liver function after MELD-based liver transplantation [IMUTECT]

Authors:

    Susanne Richter (susannerichter@gmx.net)
    Georg Polychronidis (Georg.Polychronidis@med.uni-heidelberg.de)
    Daniel N Gotthardt (Daniel.Gotthardt@med.uni-heidelberg.de)
    Philipp Houben (Philipp.Houben@med.uni-heidelberg.de)
    Thomas Giese (giese@uni-hd.de)
    Anja Sander (sander@imbi.uni-heidelberg.de)
    Colette Doerr-Harim (Colette.Doerr-Harim@med.uni-heidelberg.de)
    Markus K Diener (Markus.Diener@med.uni-heidelberg.de)
    Peter Schemmer (Peter.Schemmer@med.uni-heidelberg.de)

Version: 2
Date: 21 November 2013

Author's response to reviews:

Dear Editors,

please find enclosed our manuscript with the following revisions:

1. Competing Interests

    We specified, that Astellas was not involved in the design of the study, in the decision to submit the manuscript for publication or in the writing of the manuscript.

    Furthermore, authors declare no other conflict of interest.

2. Acknowledgements

    We acknowledged Nadya-Kaye Phillips-Houben for revising the manuscript.

Thank you very much for your consideration.

Yours sincerely,

Susanne Richter